Author's response to reviews

Title: The impact of trained patient educators on musculoskeletal clinical skills attainment in pre-clerkship medical students

Authors:

Anna E Oswald (oswald@ualberta.ca)
Mary J Bell (Mary.Bell@sunnybrook.ca)
Jeffrey Wiseman (jeffrey.wiseman@mcgill.ca)
Linda Snell (linda.snell@mcgill.ca)

Version: 5 Date: 14 September 2011

Author's response to reviews:

Dear Editors:

We are delighted to be submitting the final formatted version of our manuscript. We have relabeled the Appendices using numbers rather than letters, changed the corresponding citations in the text to match and have removed the vertical lines from the Appendices as requested. We hope this meets your satisfaction and look forward to any further comments.

Sincerely,

Dr. Anna Oswald on behalf of the study authors